Author's response to reviews

Title: Hyponatremia and hospital outcomes among patients with pneumonia: a retrospective cohort study

Authors:

Marya D Zilberberg (MZilberb@schoolph.umass.edu)
Alex Exuzides (aexuzides@ovation.org)
James Spalding (James.Spalding@US.astellas.com)
Aimee Foreman (aforeman@ovation.org)
Alison Graves Jones (agravesjones@ovation.org)
Chris Colby (colbyc@iconus.com)
Andrew F Shorr (AFShorr@DNAMail.com)

Version: 6 Date: 8 August 2008

Author's response to reviews:

Formatting changes requested

Affiliations Each affiliation should be written in full, in the following format: department, institute, city, and country. This may mean repeating institutional addresses if the authors are from different departments within an institution. Can we also ask you to link each author to his or her corresponding affiliation with the use of superscript numbers? Please do not use full capitals for country names, and please do not include full stops in abbreviated country names.

DONE

References - The reference list should contain all authors' names, if there are 30 or below. The term et al. should not be used.

DONE